Author's response to reviews

Title: Complementary and Alternative Medicine use in oncology: A questionnaire survey of patients and health care professionals

Authors:

Kah Hoong Chang (kahhoong_chang@yahoo.co.uk)
Rachel Brodie (rachel.brodie86@gmail.com)
Mei Ann Choong (meiannchoong@yahoo.co.uk)
Karl J Sweeney (karljsweeney@gmail.com)
Michael J Kerin (michael.kerin@nuigalway.ie)

Version: 4 Date: 21 April 2011

Author's response to reviews: see over
Re: Complementary and Alternative Medicine use in oncology: a questionnaire survey of patients and health care professionals

To whom it may concern,

Thank you for your comments and invitation to submit a revised manuscript. We have made changes to the format of the manuscript according to your requests. We wish to submit the formatted manuscript for publication in "BMC Cancer". We would like to thank the reviewers and editorial team for their time and consideration in reviewing our manuscript and we hope the formatting will be satisfactory.

Point-by-point descriptions of the changes made are as follows:-

Major revisions

1. Authors’ names are now written in the format of First name Initial Last name.
2. A functioning link to reference 1 is now provided.
3. Shaded areas in the table are now highlighted with bold font instead.
4. All authors are listed in the references (namely references 7, 16 and 26) instead of using ‘et al’.
5. Appendices 1 and 2 are now referred to as ‘additional files’.

Minor revisions

1. The manuscript has been proof-read.

All authors have seen and approved the revised manuscript. Should there be any further concerns which we can address please do not hesitate to contact us. We look forward to hearing your response.

Yours Sincerely

_________________________
____________________
Professor Michael Kerin  Dr Kah Hoong Chang
Corresponding Author  Principal Author
Department of Surgery  Department of Surgery
National University of Ireland, Galway  National University of Ireland, Galway
Galway, Ireland  Galway, Ireland
t: +353 91 524390  t: +353 91 544637
e: michael.kerin@nuigalway.ie  e: kahhoong_chang@yahoo.co.uk